

Geophysical Research Letters

RESEARCH LETTER

10.1029/2021GL092983

Key Points:

- Pattern effect and feedback temperature dependence lead to a proliferation of feedback definitions
- We contrast equilibrium, effective, and differential feedbacks and their physical interpretation
- Extending CMIP step-forcing simulations to 400 years would allow estimating equilibrium climate sensitivity more precisely

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

M. A. A. Rugenstein, maria.rugenstein@colostate.edu

Citation:

Rugenstein, M. A. A., & Armour, K. C. (2021). Three flavors of radiative feedbacks and their implications for estimating equilibrium climate sensitivity. *Geophysical Research Letters*, 48, e2021GL092983. https://doi. org/10.1029/2021GL092983

Received 16 FEB 2021 Accepted 3 JUN 2021 Corrected 6 SEP 2021

This article was corrected on 6 SEP 2021. See the end of the full text for details.

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

Three Flavors of Radiative Feedbacks and Their Implications for Estimating Equilibrium Climate Sensitivity

Maria A. A. Rugenstein^{1,2} ^(D) and Kyle C. Armour³ ^(D)

¹Department of Atmospheric Science, Colorado State University, Fort Collins, CO, USA, ²Max-Planck-Institute for Meteorology, Hamburg, Germany, ³School of Oceanography and Department of Atmospheric Sciences, University of Washington, Seattle, WA, USA

Abstract The realization that atmospheric radiative feedbacks depend on the underlying patterns of surface warming and global temperature, and thus, change over time has lead to a proliferation of feedback definitions and methods to estimate equilibrium climate sensitivity (ECS). We contrast three flavors of radiative feedbacks – equilibrium, effective, and differential feedback – and discuss their physical interpretations and applications. We show that their values at any given time can differ more than 1 $Wm^{-2}K^{-1}$ and their implied equilibrium or effective climate sensitivity can differ several degrees. With ten (quasi) equilibrated climate models, we show that 400 years might be enough to estimate the true ECS within a 5% error using a simple regression method utilizing the differential feedback parameter. We argue that a community-wide agreement on the interpretation of the different feedback definitions would advance the quest to narrow the estimate of climate sensitivity.

Plain Language Summary Global radiative feedbacks and their extrapolation to an equilibrium state (effective climate sensitivity) are not constant, as long assumed. This leads to the use of various methods and definitions for feedbacks and climate sensitivity. We contrast the equilibrium, effective, and differential feedback parameter definitions and discuss their implications for estimating climate sensitivity. We suggest using the differential feedback parameter, which notably does not satisfy the standard model of global energy balance, to estimate equilibrium climate sensitivity. To do so with minimum information, the climate model intercomparison protocol of simulating 150 years would need to be extended to 400 years. We further argue that a community-wide agreement on the interpretation of the feedback definitions would advance the quest to narrow the estimate of climate sensitivity and understand how feedbacks change through time.

1. Motivation

Following a radiative forcing (*F*) applied to the climate system, the degree of imbalance in the global top-of-atmosphere (TOA) radiation budget (*N*) is given by the sum of *F* and the radiative response (*R*) induced by a global surface temperature change (ΔT):

$$N = F + R. \tag{1}$$

In equilibrium, the absorbed and reflected solar shortwave and outgoing terrestrial longwave radiation must once again balance (N = 0) such that R = -F. In turn, R can be described in terms of a Taylor series expansion about a climate state (e.g., Roe, 2009):

$$R = \frac{\partial R}{\partial T} \Delta T + \mathcal{O}(\Delta T^2), \tag{2}$$

where $\tilde{\lambda} = \partial R / \partial T$ (units of Wm⁻² K⁻¹) is the radiative feedback parameter, which traditionally includes radiation anomalies associated with the Planck response and changes in lapse rate, water vapor, clouds, and surface albedo ("Charney" or fast feedbacks; Charney et al., 1979). Additional climate processes that affect R – such as changes in dust, atmospheric methane, and vegetation – are increasingly well understood and might act on similar timescales as the Charney feedbacks (e.g., Thornhill et al., 2020). The term $\mathcal{O}(\Delta T^2)$



represents the sum of all higher-order terms of the Taylor series expansion in ΔT including nonlinearities of individual processes and interactions between different feedbacks.

Two approximations are commonly applied to Equation 2: (a) that ΔT is sufficiently small that $\mathcal{O}(\Delta T^2)$ is negligible; and (b) that $\tilde{\lambda}$ is constant. Under these assumptions, and setting $\tilde{\lambda} = \lambda$, Equations 1 and 2 provide what can be considered the "Standard Model" of global climate response to forcing (popularized by Gregory et al., 2002):

$$N = F + \lambda \Delta T, \tag{3}$$

where λ must be negative for a stable climate. Using this equation, the warming at equilibrium (N = 0) can be readily expressed as $\Delta T = -F / \lambda_{eq}$, where λ_{eq} denotes the value of the net feedback that brings the system into the new equilibrium. The special case of the equilibrium response to CO₂ doubling above pre-industrial levels defines the equilibrium climate sensitivity: ECS = $-F_{2\times} / \lambda_{eq}$. Note that some studies choose an opposite sign convention for λ (sometimes referred to as α), but in either case ΔT and ECS are expressed as positive for warming.

Equation 3 has afforded substantial advances in understanding of climate change. For instance, estimates of ΔT , F, and N over the instrumental record or from proxy reconstructions of past climates provide constraints on the value of λ which, in turn, permit estimates of ECS (e.g., Sherwood et al., 2020). However, the traditional assumptions that underpin Equation 3 may not hold to the level needed to make accurate estimates of ECS. First, the $\mathcal{O}(\Delta T^2)$ term is important to account for when considering climates that are colder or warmer than the modern (e.g., Bloch-Johnson et al., 2020). Second, $\tilde{\lambda}$ (Equation 2) is in general not a constant value, and can vary with the spatial pattern of surface temperature change. This "pattern effect" (Gregory & Andrews, 2016) reflects that warming in some regions - such as the western equatorial Pacific Ocean - produce a relatively large TOA radiation response (strong negative feedback), while warming in other regions - such as the eastern tropical Pacific Ocean and Southern Ocean-produce a relatively small TOA radiation response (weak negative or positive feedback; e.g., Andrews & Webb, 2018; Ceppi & Gregory, 2017; Dong et al., 2019; Zhou et al., 2016). Furthermore, the spatial pattern of warming evolves over time owing to the different timescales of ocean adjustment in different regions, with a tendency for faster warming in more-negative feedback regions and slower warming in less-negative feedback regions leading to a trend toward less-negative values of $\tilde{\lambda}$ as climate equilibrates with an applied radiative forcing (e.g., Andrews et al., 2015; Armour et al., 2013; Dong et al., 2020; Marshall et al., 2015; Proistosescu & Huybers, 2017; Rugenstein, Caldeira, & Knutti, 2016; Senior & Mitchell, 2000; Williams et al., 2008). The pattern effect arises from changing surface warming patterns, whether those changes are caused by CO₂, aerosol, or volcanic forcing or induced by internal variability (e.g., Dessler, 2020; Gregory et al., 2020; Loeb et al., 2018; Olonscheck et al., 2020; Paynter & Frölicher, 2015).

In light of the limitations of Equation 3, it is understandable that there has been a proliferation of feedback definitions. Several reasonable choices of feedback definition can be made, including a local tangent of the radiative response defined by regressing *R* against *T* values ($\tilde{\lambda} = \partial R / \partial T$) or the use of a finite climate change of one period relative to a reference period ($\lambda = \Delta R / \Delta T$). That feedbacks can vary over time means that λ and $\tilde{\lambda}$ are not the same as λ_{eq} and that their values depend on the time period used and method by which a feedback is calculated (e.g., Barnes & Barnes, 2015; Gregory et al., 2020).

In this study, we provide physical interpretations of three different feedback definitions, discuss how they relate to each other (Section 2), and illustrate using climate model simulations when each of their applications is most appropriate (Section 3). We review proposed methods to estimate ECS from climate models and suggest how many years of a transient simulation are necessary to estimate ECS with high precision (Section 4). We argue that the quests to understand the pattern effect and to narrow the uncertainty in ECS should be accompanied by a community-wide agreement on the interpretation and calculation of feedback definitions (Section 5).



Table 1	1
---------	---

Properties of Feedback Definitions and Their Corresponding Climate Sensitivity Interpretations

-			
	Equilibrium	Effective	Differential
	feedback parameter	feedback parameter	feedback parameter

 $\lambda_{\rm eq} = -F_{2\times}/{\rm ECS}$ Difference between two equilibrated states



Dependent on the two endpoint states

- Satisfies $N = F + \lambda_{\text{eff}} \Delta T$ at N = 0, thus ECS $= -F_{2\times}/\lambda_{\text{eq}}$
- Time-invariant; not impacted by internal variability
- \rightarrow Studying $\mathcal{O}(\Delta T^2)$, i.e. nonlinearities
- \rightarrow Comparing equilibrated models and paleo-proxies

 $\lambda_{\text{eff}} = \Delta R / \Delta T$, with R = F - NChange w.r.t. a steady state



Dependent on two states (starting and current)

- Satisfies $N = F + \lambda_{\text{eff}} \Delta T$, EffCS = $-F_{2\times}/\lambda_{\text{eff}} \neq \text{ECS}$
- Can be impacted by internal variability and pattern effect, depending on period
- \rightarrow Estimate EffCS from the historical period, the paleo record, and non-equilibrated climate model simulations

 $\widetilde{\lambda} = \delta R / \delta T$ or $\widetilde{\lambda} = \delta N / \delta T$ Covariance of R (or N) and Tw.r.t. a selected state





- $N \neq F + \tilde{\lambda} \Delta T$; ECS $\neq -F_{2\times}/\tilde{\lambda}$
- Can be dominated by internal variability if δR or δT small
- $\rightarrow \ \text{Understanding causes of} \\ \text{temporal changes in } \lambda$
- \rightarrow Estimate EffCS from satellite or other short observations

Equilibrium climate sensitivity (ECS)	Effective climate sensitivity (EffCS)
Equilibrated temperature to doubled CO_2 above pre-ind. levels	Extrapolation of (linear) fit to time-evolving ΔT and R (or N)
 + Clear and concise definition - Conceptually blurs into Earth System Sensitivity with slow, non-Charney feedbacks - Requires long simulations or paleo-proxies equilibrated w.r.t. Charney feedbacks and ocean heat uptake 	 + Correlates with a model's effective and differential feedback magnitude + Used widely for short simulations and historical observations - Not well defined - Measures various degrees of equilibration, depending on input - Difficult to compare to ECS and EffCS computed with other definitions of time-frames due to missing standards on timescales, fitting methods, quantification of internal variability and F

Note. For simplicity, all values and notation in the table refer to a doubling of CO_2 , and thus, $F_{2\times}$ and ECS. "Gregory plots" illustrate a step-forcing response of a climate model in gray, the feedback parameter in red, and the climate sensitivity in orange. Noteworthy properties of the definitions are denoted with bullet points, common applications with arrows, advantages with a plus, and disadvantages with a minus.

2. Three Flavors of Radiative Feedbacks

Table 1 contrasts possible feedback definitions with illustrative sketches of the evolution of *N* with ΔT following an abrupt CO₂ increase – often referred to as a "Gregory plot" (Gregory et al., 2004). The intersect at $\Delta T = 0$ defines the effective radiative forcing, which contains tropospheric and stratospheric adjustments to the increased CO₂ concentrations before surface temperatures increase (e.g., Boucher et al., 2013) and is used as *F* in Equations 1 and 3. The intersect at N = 0 defines ΔT_{eq} , or specifically ECS in the case of a doubling of CO₂. If the N = 0 intersect is estimated by extrapolation of a regression, it is often referred to as effective climate sensitivity (EffCS). We use the expression "true ECS" and "true ΔT_{eq4x} " for doubling and quadrupling CO₂ concentrations, respectively, to indicate the actual – not estimated – equilibrium temperature value. The curvature of the gray line illustrates that climates models (here understood as general circulation models including a dynamic atmosphere and ocean; GCMs) tend to reduce the magnitude of λ with warming.

The *equilibrium feedback parameter* ($\lambda_{eq} = \Delta R_{eq} / \Delta T_{eq}$) is defined as the global radiative response per degree of surface warming calculated based on anomalies between two equilibrated climate states, notably not obtained by a regression. It can be thought of as integrating all processes which brought the climate system into the new equilibrium, which is a hypothetical state the real world will probably never experience. Providing long-enough averaging periods are used, λ_{eq} and ΔT_{eq} are constant values.

The *effective feedback parameter* ($\lambda_{\text{eff}} = \Delta R / \Delta T$) is defined as the global radiative response per degree of surface warming calculated based on anomalies between two climate states; most commonly between an initial equilibrated climate and a transient perturbation of that climate. It can be thought of as integrating all processes which brought the climate system into the new state; its extrapolation to a new equilibrium results in a measure of EffCS. This EffCS is sometimes called the "inferred" or "instantaneous" climate sensitivity, usually when calculated using observed global energy budget constraints (e.g., Armour, 2017; Lewis & Curry, 2015; Otto et al., 2013; Proistosescu & Huybers, 2017). λ_{eff} and EffCS are time-dependent as long as *N*, *F*, and ΔT change, and only become equal to the values of ΔT_{eq} and λ_{eq} as the equilibrium is approached. This results in an inherent problem: There are no commonly agreed or best-practice methods of computing values of λ_{eff} and EffCS, so care must be taken when comparing values between studies that use different periods for analysis.

Finally, and fundamentally different, is the *differential feedback parameter*, ($\tilde{\lambda} = \delta R / \delta T$), calculated based on $R - \Delta T$ regression (or $N - \Delta T$ regression in the case the forcing is constant). In a step-forcing GCM experiment, δR and δT can be a few Wm⁻² and K, respectively, representing a forced response or a radiative feedback in the traditional sense (Andrews et al., 2015; Ceppi & Gregory, 2019; Knutti & Rugenstein, 2015). In observations and simulations with continuously changing or constant forcing, δR and δT might cover only a few tenth of Wm⁻² and K, respectively and referred to as radiative restoration strength (e.g., Colman & Hanson, 2017; Donohoe et al., 2014; Lutsko & Takahashi, 2018; Murphy, 2010; Proistosescu et al., 2018).

 $\tilde{\lambda}$ represents all the processes causing δR to change with δT for the period over which the calculation is performed, as opposed to λ_{eq} and λ_{eff} representing the integrated nature of ΔR per ΔT between two states. For example, under transient warming, $\tilde{\lambda}$ quantifies changes in *R* per δT associated with the regions which are warming during that time (Andrews et al., 2015; Armour et al., 2013; Rugenstein, Caldeira, et al., 2016).

Importantly, $\tilde{\lambda}$ does not satisfy the equation $N = F + \lambda \Delta T$, unless a radical re-definition of the forcing term is used such as a "virtual forcing" (Rugenstein, Gregory, et al., 2016; Williams et al., 2008). Thus, while a value $-F_{2\times} / \tilde{\lambda}$ can be readily calculated it should not be interpreted as ECS. However, the regression $\delta R / \delta T$ extrapolated to N = 0 does result in an EffCS – the use of which we will demonstrate below. Thus, when the terminology EffCS is used, care must be taken to determine whether the calculation was performed using the *effective* or *differential* feedback definition.

Importantly, the different feedback definitions and resulting climate sensitivities can be only meaningfully compared if their differences are understood and accounted for. First attempts of developing relations between climate sensitivity estimates or radiative feedbacks driven by different surface warming patterns ("transfer functions") have been made (Armour, 2017; Andrews et al., 2018; Sherwood et al., 2020). Refined transfer functions require an improved process understanding, especially on the influence of large spatial warming patterns on TOA fluxes in the observed record, climate models, and the paleo record.

3. Radiative Feedbacks and Their Implications for Estimating ECS in GCMs

The time evolution of the effective and differential feedbacks differs substantially in simulations with an idealized step-forcing and simulations with slowly or realistically increases of CO_2 concentrations. For illustrative purposes, we use simulations of one model, CESM 1.0.4, forced with a step-forcing of quadrupling



Figure 1. Flavors of feedback parameters in the model CESM 1.0.4: Gregory plots for a CO_2 step-forcing simulation (a) and an ensemble of 10 simulations with slowly increasing CO_2 concentrations (b), their evolution of effective and differential feedback parameters (c and d), and the implied estimate of the equilibrium temperature ΔT_{eq4x} (e and f). In panel (a), the gray dots around (0,0) indicate annual means of the 1000-year long control simulation; the colored lines indicate, for illustrative purposes, the regressions that determine the feedback parameters shown in panel c around year 1,000. Years 900–1,100 are indicated in lighter gray and the temperature ranges for the regressions of the differential feedback parameters in green and blue. In panel (b), black dots indicate the ensemble average and light gray dots year 130–150 values to highlight the time of CO_2 quadrupling. The feedback parameters shown in panel d are calculated according to the equations in the text (in time and not temperature space) and thus not indicated in panel b. In panel d and f, each ensemble member is depicted as thin and the ensemble average as darker thick line. In panel c and d the uncertainty bars for $\tilde{\lambda}_{eql}$ (also shown in panel a) indicate the 2.5%–97.5% confidence interval. M values in brackets refer to methods in Table 2.

CO₂ concentrations ("abrupt4x") and with an annually increasing CO₂ concentration of 1% ("1pct"). Here and later, N and ΔT are the global and annual mean anomalies of these simulations relative to a stable control simulation averaged over 1,000 years. In a step-forcing simulation (Figure 1a), the equilibrium feedback parameter is calculated as $\lambda_{eq4x} = -F_{fixedSST4x} / \Delta T_{eq4x}$, where $F_{fixedSST4x}$ is obtained from a dedicated simulation of quadrupling CO₂ concentrations while holding the sea surface temperatures and sea ice (but not the land temperatures) fixed at pre-industrial values (7.25 Wm^{-2}). ΔT_{eq} is the average temperature over several centuries at the end of the 5,300 years long simulation. The effective feedback parameter is calculated as $\lambda_{eff4x} = -(F_{fixedSST4x} - N(t)) / \Delta T(t)$, where N(t) and $\Delta T(t)$ are the 30-year moving averages of the respective terms. λ_{eff4x} changes most initially and then slowly approaches λ_{eq4x} . We calculate two *differential feedback parameters:* $\tilde{\lambda}_{4_{XA}}$ is the regression of all N on ΔT values within a temperature bin of 1.5 K which is slid through the entire ΔT range of 1–6.8 K covered by the simulation. $\tilde{\lambda}_{4xB}$ is the slope of the regression of all N and ΔT values prior to each time step, e.g., at year 100 regressing years 1–100, at year 101 regressing years 1–101, etc.. It changes more slowly than $\lambda_{4_{XA}}$ over the course of several thousand years (Figure 1c). Finally, λ_{ctl} is the radiative restoration strength of an unperturbed, 1,000 years long control climate (Figure 1a), obtained by regressing annual deviations (spanning 2.5 Wm^{-2}) of the mean N onto the deviations (spanning 0.87 K), and is influenced by correlated noise (Gregory et al., 2020; Proistosescu et al., 2018).

In a simulation with slowly increasing forcing (Figure 1b), the *effective feedback parameter*, $\lambda_{eff1pct}$ is calculated as $(F_{1pct}(t) - N(t)) / \Delta T(t)$, with N(t) and $\Delta T(t)$ being the 30-year moving averages of the respective time series. $F_{1pct}(t) = F_{fixedSST4x} t / 140$, with t referring to time in years, is the radiative forcing assuming linearity with the logarithm of CO₂ concentration. The *differential feedback parameter* ($\tilde{\lambda}_{1pct}$) is calculated as the linear regression of $F_{1pct}(t) - N(t)$ onto $\Delta T(t)$ for 30-year bins slid through the time series (as opposed to using bins defined in temperature space as for the *abrupt4x* simulation). With stronger forcing and more warming, internal variability – illustrated by 10 ensemble members branched-off different years of the control simulations – becomes negligible after 120 years for $\lambda_{eff1pct}$, but not for $\tilde{\lambda}_{1pct}$ (Figure 1d).

The choices of feedback definitions imply different estimates of ΔT_{eq4x} (Figure 1e and 1f): In a step forcing simulation, on the one hand, $\Delta T_{eq4x} = -F_{fixedSST4x} / \lambda_{eff4x}(t)$ or extrapolating the regression used to define $\tilde{\lambda}_{4xB}(t)$ to N = 0 results in a similar time-evolving ΔT_{eq4x} estimate which approaches the true ΔT_{eq4x} within several millennia. On the other hand, extrapolating the regression used to define $\tilde{\lambda}_{4xA}(t)$ to N = 0 results in a close estimate of ΔT_{eq4x} within 200 years. In the simulations with slower changing forcing, using $\Delta T_{eq4x} = -F_{fixedSST4x} / \lambda_{eff1pct}(t)$ or $\Delta T_{eq4x} = -F_{fixedSST4x} / \tilde{\lambda}_{1pct}(t)$ leads to arbitrary, unreasonable estimates of ΔT_{eq4x} . $\lambda_{eff1pct}(t)$ underestimates ΔT_{eq4x} when it passes the equivalent forcing value around year 140, and hits the correct ΔT_{eq4x} value coincidentally after 240 years at a forcing of $\approx 12 \text{ Wm}^{-2}$. $\tilde{\lambda}_{1pct}(t)$ is influenced by internal variability throughout the simulation but passes $\Delta T_{eq4x} = 6.8$ K at the correct forcing values around year 140 (see discussion in Gregory et al., 2015). The values of $\lambda_{eff1pct}$ and $\tilde{\lambda}_{1pct}$ strongly depend on the formulation of $F_{1pct}(t)$ for high forcing levels (e.g., Bloch-Johnson et al., 2020; Byrne & Goldblatt, 2014; Gregory et al., 2015).

4. Estimating Climate Sensitivities From Climate Models

Since the wide recognition of the inconstancy of radiative feedbacks many methods have been proposed to estimate the true ECS (labeled M1 to M11 in Table 2). Recently, a large number of GCMs were integrated to near or full equilibrium (1,000–6,000 years simulation time; LongRunMIP, Rugenstein et al., 2019) which is computationally expensive, and thus not frequently done. Rugenstein et al. (2020) showed that across 14 models the most commonly used extrapolation methods (M1, M2, and M7) underestimate the true ΔT_{eq4x} or true ECS (or where not available, the estimate using M5) by 5%–20%. Dunne et al. (2020) showed that M10 underestimates a "long ECS" (defined by a variation of M6) by 6% ± 7% in 13 models and suggested another variation of M6 – regressing 5-year averages between year 50 and 150 – to most closely estimate "long ECS." The general underestimation of the true ECS has been also shown for single models and estimation methods (e.g., Danabasoglu & Gent, 2009; Li et al., 2013; Paynter et al., 2018; Saint-Martin et al., 2019; Senior & Mitchell, 2000; Winton et al., 2020).



Table 2 ECS Estimation Methods, all are Variants of Effective Climate Sensitivity					
Method #	Explanation	Example references			
M1*	linearly regressing year 1–150 of N against ΔT	Andrews et al. (2012); Gregory et al. (2004); Sherwood et al. (2020)			
M2*	linearly regressing year 20 or 21–150 of N against ΔT	Armour (2017); Andrews et al. (2015); Forster (2016)			
M3*	local tangent, $\delta N / \delta \Delta T$, to the $N - \Delta T$ point cloud using a sliding window of ΔT large enough to capture the warming response	Ceppi and Gregory (2019); Knutti and Rugenstein (2015); Rugenstein, Gregory, et al. (2016)			
M4	differencing either $(F - N(t)) / \Delta T(t)$ for an increasing number of years or $(N_2 - N_1) / (\Delta T_2 - \Delta T_1)$ for two time periods	Dong et al. (2019); Rugenstein et al. (2020); Senior and Mitchell (2000)			
M5*	linearly regression all years of N against ΔT within the last 15% of warming of a simulation which is at least 1,000 years long	Rugenstein et al. (2020) and this section; For equilibrated simulations this is equivalent to averaging the temperature over the final decades.			
M6*	linearly regressing five, ten,, 50-year averages of N against ΔT	Dunne et al. (2020); Gregory et al. (2004); Winton et al. (2020)			
M7	fitting an energy balance model including ocean heat uptake efficacy, EBM- ϵ , to a number of years in N against ΔT	Dai et al. (2020); Geoffroy et al. (2013); Held et al. (2010); Rohrschneider et al. (2019)			
M8*	eigenmode decomposition, fitting exponential functions through time series of ΔT (and <i>N</i>) with conditions about their common timescales or axes intersects	Caldeira and Myhrvold (2013); Mauritsen et al. (2018); Proistosescu and Huybers (2017); Sanderson (2019)			
M9	Cess-sensitivity: differencing two equilibrated atmospheric states forced with different SST $(\Delta N / \Delta T)$, <i>F</i> assumptions come from other simulations	Becker and Wing (2020); Cess and Potter (1988); Gettelman et al. (2012); Ringer et al. (2014)			
M10	slab ocean simulations, which prescribe a spatially varying ocean mixed layer depth and heat uptake distribution	Danabasoglu and Gent (2009); Dunne et al. (2020)			
M11	a series of specific, short, coupled GCM simulations with different forcing levels	Saint-Martin et al. (2019)			

Note. In the first column, methods implying a differential versus an effective feedback parameter definition are indicated with versus without an asterisk. Abbreviations: GCM, general circulation model; SST, sea surface temperature.

Figure 2 summarizes and expands these analyses in that it compares estimates of ΔT_{eq4x} using (variations of) methods M1–M8. The computation of M1–M5 follows the description in Table 2. For M6, we show two versions of linearly regressing *N* against ΔT : of the averages of years 50–100, 101–150, 151–200, 201–250, and 251–300 (Winton et al., 2020) and of the 5-year averages of year 50–150 (Dunne et al., 2020). For M7, the energy balance model formulation of Geoffroy et al. (2013) is fitted to ΔT and *N* of years 1–150 and years 1–1,000. Likewise, for M8 the Bayesian framework of Proistosescu and Huybers (2017) is used to fit exponential functions with three exponents to years 1–150 and years 1–1,000 of the ΔT and *N* timeseries. Using 1,000 years for method M7 and M8 somewhat defeats the purpose of estimating ΔT_{eq4x} as the simulation is close to the new equilibrium then. We cannot evaluate M9–M11 because these methods require dedicated climate model simulations not done routinely and not available in the LongRunMIP archive.

Figure 2 shows that different methods to estimate ΔT_{eq4x} or ECS qualitatively measure the overall sensitivity of a model in that more sensitive models appear more sensitive and less sensitive models appear less sensitive, independent on the exact method. However, the relative rank of the models does not stay the same for different methods, especially if the overall sensitivity is similar or the sensitivity is very high. Furthermore – in our set of models – more sensitive models have a higher spread in sensitivity measures than less sensitive

10.1029/2021GL092983

Geophysical Research Letters





Figure 2. Values of ΔT_{eq4x} estimates using variations of method M1–M8 for 10 general circulation models (GCMs) *abrupt4x* simulations. The correlation coefficient between each method and the true ΔT_{eq4x} quantified by M5 is indicated as a bold number in brackets in the legend. The simulation length and *N* averaged over the last 50 years are mentioned beneath the model's name. "Range of local tangents" span the minimum to maximum estimates of ΔT_{eq4x} based on the differential feedback parameter calculated by regressing N onto Δ T values.

models. Feedback temperature dependence would increase that tendency. Finally, the different methods explain the evolution of the surface temperature across models differently well through time (Figure S1).

We quantify how many years of a transient simulation are necessary to predict the true ΔT_{eq4x} correctly using the differential feedback definition and extrapolating the linear regression to equilibrium (M1–M5). Figure 3 shows the model-mean fraction of ΔT_{eq4x} depending on first and last year of a linear regression of *N* on ΔT and reveals three insights: (a) Solely increasing the number of years in the linear regression (M4) requires many years to approach the true ΔT_{eq4x} (zoom-in panel). Thus, skipping several decades in the beginning of the simulation (as in M2, M3, M6) is necessary to gain predictive skill. (b) Regressing too few years too far into the simulation will mostly sample internal variability and randomly predict ΔT_{eq4x} values more than 20% too high or too low. (c) The standard deviation between the 10 models is large and only reduces on the centennial timescale. We suggest to linearly regress years 100–400 to estimate ΔT_{eq4x}



within a 5% error (see also Figure 2). For single models, regressing fewer years results in the same accuracy, however, for generally more sensitive models even more years might be necessary, which is likely also true for other CMIP6 models (e.g., Forster et al., 2019; Tokarska et al., 2020; Zelinka et al., 2020). Supporting Information S1 and Figures S2 and S3 discuss the equilibration pace for the subset of five models for which *abrupt2x* and *abrupt4x* simulations are available.

5. Conclusions

We have shown that the equilibrium, effective, and differential feedback parameter definitions are physically meaningful (Table 1), however, they represent different properties of the climate system, are not easily translatable, and differ in value and implications for estimating ECS (Figure 1). The realization that feedbacks change in response to surface warming patterns and global mean temperature has led to a development of many methods to estimate ECS (Table 2), relying on different feedback definitions. All methods capture the overall sensitivity of the models but can result in ECS or ΔT_{eq4x} estimates that differ more than 1 K and do not necessarily record the equilibrated state (Figure 2).

If the goal is to predict ECS or ΔT_{eq4x} from a transient simulation as quickly and precisely as possible, the first couple of decades must be discarded when applying regression methods (Figure 3; Dai et al., 2020; Dunne et al., 2020; Rugenstein et al., 2020; Winton et al., 2020). Using the differential feedback definition and extrapolating the linear regression of annual averages of *N* against ΔT of years 100–400 of an *abrupt4x* simulation results in a ΔT_{eq4x} estimate with less than 5% error across the 10 climate models for which true equilibrium values are available. Thus, we suggest an extension of the CMIP6 step-forcing simulations of *abrupt4x* and/or *abrupt2x* in the CMIP6 protocols of DECK and nonlinMIP (Eyring et al., 2016; Good et al., 2016) from their typical length of 150 years to a length of 400 years.



Figure 3. Model-mean fraction of ΔT_{eq4x} quantified by M5 depending on first and last year of a linear regression of N on ΔT (color shading) for the same ten *abrupt4x* simulations as in Figure 2. The standard deviation across the simulations is shown in hashed patterns; for example, a regression of years 100–220 results in a 5% error for the model mean (gray shading), but the standard deviation across the 10 models is still 10% (line hashing). The white dot indicates the suggested method of regressing years 100–400. The axes of the inset are first and last year of the regression as in the larger plot.

While the use of ECS has some distinct advantages over effective or inferred climate sensitivities because no adjustment for the degree of equilibration or the surface warming patterns is necessary, characterizing instead feedbacks and their uncertainty for a specific climate state might be methodologically preferred (Dunne et al., 2020; Klein et al., 2017; Roe, 2009). The most pressing open questions are (a) how to relate feedback estimates from different states of surface warming patterns or global mean temperatures, thus, developing feedback transfer functions based on physical processes; (b) how to apply the global energy balance framework to short observations and short and limited-domain very high resolution model simulations which are dominated by internal variability or representing one specific state which may or may not relate to anticipated future warming; (c) how to re-phrase the global energy balance model to include the pattern effect; (d) how to treat non-Charney feedbacks present in Earth System Models and the real world on decadal to millennial timescales in the energy balance perspective.

We conclude that the quests to understand the pattern effect, to estimate ECS from observations, and to narrow the overall uncertainty in ECS with process studies and paleo climate estimates should be accompanied by a community-wide agreement on the interpretation of the different feedback definitions and the methods to compute them.

Data Availability Statement

All data behind Figures 1–3 are available under: http://dx.doi.org/10.25675/10217/232176.

References

- Andrews, T., Gregory, J. M., Paynter, D., Silvers, L. G., Zhou, C., Mauritsen, T., et al. (2018). Accounting for changing temperature patterns increases historical estimates of climate sensitivity. *Geophysical Research Letters*, 45(16), 8490–8499. https://doi.org/10.1029/2018gl078887
 Andrews, T., Gregory, J. M., & Webb, M. J. (2015). The dependence of radiative forcing and feedback on evolving patterns of surface tem-
- perature change in climate models. *Journal of Climate*, 28(4), 1630–1648. https://doi.org/10.1175/JCLI-D-14-00545.1
- Andrews, T., Gregory, J. M., Webb, M. J., & Taylor, K. E. (2012). Forcing, feedbacks and climate sensitivity in CMIP5 coupled atmosphere-ocean climate models. *Geophysical Research Letters*, 39(9), L09712. https://doi.org/10.1029/2012GL051607
- Andrews, T., & Webb, M. J. (2018). The dependence of global cloud and lapse rate feedbacks on the spatial structure of tropical pacific warming. *Journal of Climate*, *31*(2), 641–654. https://doi.org/10.1175/JCLI-D-17-0087.1
- Armour, K. C. (2017). Energy budget constraints on climate sensitivity in light of inconstant climate feedbacks. *Nature Climate Change*, 7, 331–335. https://doi.org/10.1038/nclimate3278
- Armour, K. C., Bitz, C. M., & Roe, G. H. (2013). Time-varying climate sensitivity from regional feedbacks. Journal of Climate, 26(13), 4518–4534. https://doi.org/10.1175/JCLI-D-12-00544.1
- Barnes, E. A., & Barnes, R. J. (2015). Estimating linear trends: Simple linear regression versus epoch differences. Journal of Climate, 28(24), 9969–9976. https://doi.org/10.1175/jcli-d-15-0032.1
- Becker, T., & Wing, A. A. (2020). Understanding the extreme spread in climate sensitivity within the radiative-convective equilibrium model intercomparison project. *Journal of Advances in Modeling Earth Systems*, *12*(10), e2020MS002165. https://doi.org/10.1029/2020ms002165
- Bloch-Johnson, J., Rugenstein, M., Stolpe, M. B., Rohrschneider, T., Zheng, Y., & Gregory, J. (2020). Climate sensitivity increases under higher CO₂ levels due to feedback temperature dependence. *Geophysical Research Letters*, 48, e2020GL089074. https://doi. org/10.1029/2020GL089074
- Boucher, O., Randall, D., Artaxo, P., Bretherton, C., Feingold, G., Forster, P. M., et al. (2013). Clouds and Aerosols. In Climate Change 2013: The Physical Science Basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, UK; New York, NY: Cambridge University Press.
- Byrne, B., & Goldblatt, C. (2014). Radiative forcing at high concentrations of well-mixed greenhouse gases. *Geophysical Research Letters*, 41, 152–160. https://doi.org/10.1002/2013gl058456
- Caldeira, K., & Myhrvold, N. P. (2013). Projections of the pace of warming following an abrupt increase in atmospheric carbon dioxide concentration. *Environmental Research Letters*, 8(3), 034039. https://doi.org/10.1088/1748-9326/8/3/034039
- Ceppi, P., & Gregory, J. M. (2017). Relationship of tropospheric stability to climate sensitivity and Earth's observed radiation budget. *Proceedings of the National Academy of Sciences*, 114(50), 13126–13131. https://doi.org/10.1073/pnas.1714308114
- Ceppi, P., & Gregory, J. M. (2019). A refined model for the Earth's global energy balance. *Climate Dynamics*, 53, 4781–4797. https://doi.org/10.1007/s00382-019-04825-x
- Cess, R. D., & Potter, G. L. (1988). A methodology for understanding and intercomparing atmospheric climate feedback processes in general circulation models. *Journal of Geophysical Research*, 93(D7), 8305–8314. https://doi.org/10.1029/jd093id07p08305
- Charney, J., Arakawa, A., Baker, D., Bolin, B., Dickinson, R. E., Goody, R., et al. (1979). Carbon Dioxide and Climate: A Scientific Assessment (Tech. Rep.). Washington, DC. National Academy of Science.
- Colman, R., & Hanson, L. (2017). On the relative strength of radiative feedbacks under climate variability and change. *Climate Dynamics*, 49(5), 2115–2129. https://doi.org/10.1007/s00382-016-3441-8
- Dai, A., Huang, D., Rose, B. E. J., Zhu, J., & Tian, X. (2020). Improved methods for estimating equilibrium climate sensitivity from transient warming simulations. *Climate Dynamics*. https://doi.org/10.1007/s00382-020-05242-1
- Danabasoglu, G., & Gent, P. R. (2009). Equilibrium climate sensitivity: Is it accurate to use a slab ocean model? *Journal of Climate*, 22(9), 2494–2499. https://doi.org/10.1175/2008JCLI2596.1
- Dessler, A. E. (2020). Potential problems measuring climate sensitivity from the historical record. *Journal of Climate*, 33(6), 2237–2248. https://doi.org/10.1175/JCLI-D-19-0476.1

Acknowledgments

The authors thank Jonathan Gregory for the idea for Figure 3 and enlightening discussions. The authors acknowledge Jan-Moritz Witt for calculating and Cristian Proistosescu for developing the Bayesian "3-exp fits." The authors thank Jochem Marotzke, Bjorn Stevens, and two anonymous reviewers for comments on the manuscript and subjects therein. K.C. Armour was supported by National Science Foundation Grant AGS-1752796. M. Rugenstein was supported by the Humboldt Foundation. Open access funding enabled and organized by Projekt DEAL.



Dong, Y., Armour, K. C., Zelinka, M. D., Proistosescu, C., Battisti, D. S., Zhou, C., & Andrews, T. (2020). Intermodel spread in the pattern effect and its contribution to climate sensitivity in CMIP5 and CMIP6 Models. *Journal of Climate*, 33(18), 7755–7775. https://doi. org/10.1175/JCLI-D-19-1011.1

Dong, Y., Proistosescu, C., Armour, K. C., & Battisti, D. S. (2019). Attributing historical and future evolution of radiative feedbacks to regional warming patterns using a green's function approach: The preeminence of the western pacific. *Journal of Climate*, 32(17), 5471–5491. https://doi.org/10.1175/JCLI-D-18-0843.1

- Donohoe, A., Armour, K. C., Pendergrass, A. G., & Battisti, D. S. (2014). Shortwave and longwave radiative contributions to global warming under increasing CO₂. Proceedings of the National Academy of Sciences, 111(47), 16700–16705. https://doi.org/10.1073/pnas.1412190111
- Dunne, J. P., Winton, M., Bacmeister, J., Danabasoglu, G., Gettelman, A., Golaz, J.-C., et al. (2020). Comparison of equilibrium climate sensitivity estimates from Slab Ocean, 150-Year, and longer simulations. *Geophysical Research Letters*, 47, e2020GL088852. https://doi. org/10.1029/2020GL088852
- Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the coupled model intercomparison Project Phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development*, 9(5), 1937–1958. https://doi. org/10.5194/gmd-9-1937-2016
- Forster, P. M. (2016). Inference of climate sensitivity from analysis of Earth's energy budget. *Annual Review of Earth and Planetary Sciences*, 44(1), 85–106. https://doi.org/10.1146/annurev-earth-060614-105156
- Forster, P. M., Maycock, A. C., McKenna, C. M., & Smith, C. J. (2019). Latest climate models confirm need for urgent mitigation. Nature Climate Change, 10, 7–10. https://doi.org/10.1038/s41558-019-0660-0
- Geoffroy, O., Saint-Martin, D., Bellon, G., Voldoire, A., Olivié, D., & Tytéca, S. (2013). Transient climate response in a two-layer energy-balance model. Part II: Representation of the efficacy of deep-ocean heat uptake and validation for CMIP5 AOGCMs. *Journal of Climate*, 26(6), 1859–1876. https://doi.org/10.1175/JCLI-D-12-00196.1
- Gettelman, A., Kay, J. E., & Shell, K. M. (2012). The evolution of climate sensitivity and climate feedbacks in the community atmosphere model. *Journal of Climate*, 25(5), 1453–1469. https://doi.org/10.1175/JCLI-D-11-00197.1
- Good, P., Andrews, T., Chadwick, R., Dufresne, J.-L., Gregory, J. M., Lowe, J. A., et al. (2016). nonlinMIP contribution to CMIP6: Model intercomparison project for non-linear mechanisms: Physical basis, experimental design and analysis principles (v1.0). Geoscientific Model Development, 9(11), 4019–4028. https://doi.org/10.5194/gmd-9-4019-2016
- Gregory, J. M., & Andrews, T. (2016). Variation in climate sensitivity and feedback parameters during the historical period. *Geophysical Research Letters*, 43, 3911–3920. https://doi.org/10.1002/2016GL068406
- Gregory, J. M., Andrews, T., Ceppi, P., Mauritsen, T., & Webb, M. J. (2020). How accurately can the climate sensitivity to co₂ be estimated from historical climate change? *Climate Dynamics*, 54(1), 129–157. https://doi.org/10.1007/s00382-019-04991-y
- Gregory, J. M., Andrews, T., & Good, P. (2015). The inconstancy of the transient climate response parameter under increasing CO₂. Philosophical Transactions of the Royal Society of London A: Mathematical, Physical and Engineering Sciences, 373. https://doi.org/10.1098/ rsta.2014.0417
- Gregory, J. M., Ingram, W. J., Palmer, M. A., Jones, G. S., Stott, P. A., Thorpe, R. B., et al. (2004). A new method for diagnosing radiative forcing and climate sensitivity. *Geophysical Research Letters*, 31, L03205. https://doi.org/10.1029/2003GL018747
- Gregory, J. M., Stouffer, R. J., Raper, S. C. B., Stott, P. A., & Rayner, S. A. (2002). An observationally based estimate of the climate sensitivity. Journal of Climate, 15(22), 3117–3121. https://doi.org/10.1175/1520-0442(2002)015<3117:AOBEOT>2.0.CO;2
- Held, I., Winton, M., Takahashi, K., Delworth, T. L., Zeng, F., & Vallis, G. (2010). Probing the fast and slow components of global warming by returning abruptly to preindustrial forcing. *Journal of Climate*, 23, 2418–2427. https://doi.org/10.1175/2009jcli3466.1
- Klein, S. A., Hall, A., Norris, J. R., & Pincus, R. (2017). Low-cloud feedbacks from cloud-controlling factors: A review. Surveys in Geophysics. https://doi.org/10.1007/s10712-017-9433-3
- Knutti, R., & Rugenstein, M. A. A. (2015). Feedbacks, climate sensitivity and the limits of linear models. Philosophical Transactions of the Royal Society of London A: Mathematical, Physical and Engineering Sciences, 373. https://doi.org/10.1098/rsta.2015.0146
- Lewis, N., & Curry, J. A. (2015). The implications for climate sensitivity of AR5 forcing and heat uptake estimates. *Climate Dynamics*, 45(3), 1009–1023. https://doi.org/10.1007/s00382-014-2342-y
- Li, C., Notz, D., Tietsche, S., & Marotzke, J. (2013). The transient versus the equilibrium response of sea ice to global warming. *Journal of Climate*, 26(15), 5624–5636. https://doi.org/10.1175/JCLI-D-12-00492.1
- Loeb, N. G., Doelling, D. R., Wang, H., Su, W., Nguyen, C., Corbett, J. G., et al. (2018). Clouds and the Earth's Radiant Energy System (CERES) Energy Balanced and Filled (EBAF) Top-of-Atmosphere (TOA) Edition-4.0 Data Product. *Journal of Climate*, 31(2), 895–918. https://doi.org/10.1175/JCLI-D-17-0208.1
- Lutsko, N. J., & Takahashi, K. (2018). What can the internal variability of CMIP5 models tell us about their climate sensitivity? Journal of Climate, 31(13), 5051–5069. https://doi.org/10.1175/JCLI-D-17-0736.1
- Marshall, J., Scott, J. R., Armour, K. C., Campin, J. M., Kelley, M., & Romanou, A. (2015). The ocean's role in the transient response of climate to abrupt greenhouse gas forcing. *Climate Dynamics*, 44(7), 2287–2299. https://doi.org/10.1007/s00382-014-2308-0
- Mauritsen, T., Bader, J., Becker, T., Behrens, J., Bittner, M., Brokopf, R., et al. (2018). Developments in the MPI-M Earth System Model version 1.2 (MPI-ESM 1.2) and its response to increasing CO₂. Journal of Advances in Modeling Earth Systems. https://doi. org/10.1029/2018MS001400

Murphy, D. M. (2010). Constraining climate sensitivity with linear fits to outgoing radiation. *Geophysical Research Letters*, 37, L09704. https://doi.org/10.1029/2010GL042911

- Olonscheck, D., Rugenstein, M., & Marotzke, J. (2020). Broad consistency between observed and simulated trends in sea surface temperature patterns. *Geophysical Research Letters*, 47, e2019GL086773. https://doi.org/10.1029/2019GL086773
- Otto, A., Otto, F. E. L., Boucher, O., Church, J., Hegerl, G., Forster, P. M., et al. (2013). Energy budget constraints on climate response. *Nature Geoscience*, 6(6), 415–416. https://doi.org/10.1038/ngeo1836
- Paynter, D., & Frölicher, T. L. (2015). Sensitivity of radiative forcing, ocean heat uptake, and climate feedback to changes in anthropogenic greenhouse gases and aerosols. Journal of Geophysical Research: Atmospheres, 120, 9837–9854. https://doi.org/10.1002/2015JD023364
- Paynter, D., Frölicher, T. L., Horowitz, L. W., & Silvers, L. G. (2018). Equilibrium climate sensitivity obtained from multimillennial runs of two GFDL climate models. *Journal of Geophysical Research: Atmospheres, 123*(4), 1921–1941. https://doi.org/10.1002/2017jd027885
- Proistosescu, C., Donohoe, A., Armour, K. C., Roe, G. H., Stuecker, M. F., & Bitz, C. M. (2018). Radiative feedbacks from stochastic variability in surface temperature and radiative imbalance. *Geophysical Research Letters*, 45, 5082–5094. https://doi.org/10.1029/2018gl077678
- Proistosescu, C., & Huybers, P. J. (2017). Slow climate mode reconciles historical and model-based estimates of climate sensitivity. *Science Advances*, 3(7). https://doi.org/10.1126/sciadv.1602821



- Ringer, M. A., Andrews, T., & Webb, M. J. (2014). Global-mean radiative feedbacks and forcing in atmosphere-only and coupled atmosphere-ocean climate change experiments. *Geophysical Research Letters*, 41, 4035–4042. https://doi.org/10.1002/2014GL060347
- Roe, G. (2009). Feedbacks, timescales, and seeing red. Annual Review of Earth and Planetary Sciences, 37(1), 93–115. https://doi.org/10.1146/annurev.earth.061008.134734
- Rohrschneider, T., Stevens, B., & Mauritsen, T. (2019). On simple representations of the climate response to external radiative forcing. *Climate Dynamics*. https://doi.org/10.1007/s00382-019-04686-4
- Rugenstein, M., Bloch-Johnson, J., Abe-Ouchi, A., Andrews, T., Beyerle, U., Cao, L., et al. (2019). LongRunMIP: Motivation and design for a large collection of millennial-length AOGCM simulations. *Bulletin of the American Meteorological Society*, 100(12), 2551–2570. https://doi.org/10.1175/BAMS-D-19-0068.1
- Rugenstein, M., Bloch-Johnson, J., Gregory, J., Andrews, T., Mauritsen, T., Li, C., et al. (2020). Equilibrium climate sensitivity estimated by equilibrating climate models. *Geophysical Research Letters*, 47, e2019GL083898. https://doi.org/10.1029/2019GL083898
- Rugenstein, M. A. A., Caldeira, K., & Knutti, R. (2016). Dependence of global radiative feedbacks on evolving patterns of surface heat fluxes. *Geophysical Research Letters*, 43, 9877–9885. https://doi.org/10.1002/2016GL070907
- Rugenstein, M. A. A., Gregory, J. M., Schaller, N., Sedláček, J., & Knutti, R. (2016). Multiannual Ocean–Atmosphere Adjustments to Radiative Forcing. Journal of Climate, 29(15), 5643–5659. https://doi.org/10.1175/JCLI-D-16-0312.1
- Saint-Martin, D., Geoffroy, O., Watson, L., Douville, H., Bellon, G., Voldoire, A., et al. (2019). Fast forward to perturbed equilibrium climate. *Geophysical Research Letters*. https://doi.org/10.1029/2019GL083031
- Sanderson, B. (2019). Relating climate sensitivity indices to projection uncertainty. Earth System Dynamics Discussions, 2019, 1–18. https:// doi.org/10.5194/esd-2019-77
- Senior, C. A., & Mitchell, J. F. B. (2000). The time-dependence of climate sensitivity. *Geophysical Research Letters*, 27(17), 2685–2688. https://doi.org/10.1029/2000GL011373
- Sherwood, S., Webb, M. J., Annan, J. D., Armour, K. C., Forster, P. M., Hargreaves, J. C., et al. (2020). An assessment of Earth's climate sensitivity using multiple lines of evidence. *Reviews of Geophysics*. https://doi.org/10.1029/2019RG000678
- Thornhill, G., Collins, W., Olivié, D., Archibald, A., Bauer, S., Checa-Garcia, R., et al. (2020). Climate-driven chemistry and aerosol feedbacks in CMIP6 Earth system models. *Atmospheric Chemistry and Physics Discussions*, 2020, 1–36. https://doi.org/10.5194/acp-2019-1207 Tokarska, K., Stolpe, M., Sippel, S., Fischer, E., Smith, C., Lehner, F., & Knutti, R. (2020). Past warming trend constrains future warming in
- CMIP6 models. Science Advances. https://doi.org/10.1126/sciadv.aaz9549
- Williams, K. D., Ingram, W. J., & Gregory, J. M. (2008). Time variation of effective climate sensitivity in GCMs. Journal of Climate, 21(19), 5076–5090. https://doi.org/10.1175/2008JCLI2371.1
- Winton, M., Adcroft, A., Dunne, J. P., Held, I. M., Shevliakova, E., Zhao, M., et al. (2020). Climate Sensitivity of GFDL's CM4.0. Journal of Advances in Modeling Earth Systems, 12(1). https://doi.org/10.1029/2019MS001838
- Zelinka, M. D., Myers, T. A., McCoy, D. T., Po-Chedley, S., Caldwell, P. M., Ceppi, P., et al. (2020). Causes of higher climate sensitivity in CMIP6 Models. *Geophysical Research Letters*, 47, e2019GL085782. https://doi.org/10.1029/2019GL085782
- Zhou, C., Zelinka, M. D., & Klein, S. A. (2016). Impact of decadal cloud variations on the Earth's energy budget. *Nature Geoscience*, 9(12), 871–874. https://doi.org/10.1038/ngeo2828

Erratum

A typographical error was corrected. This version may be considered the authoritative version of record.